Use of selective gut decontamination in critically ill children: PICnIC a pilot RCT and mixedmethods study

Alanna Brown,¹ Paloma Ferrando-Vivas,¹ Mariana Popa,² Gema Milla de la Fuente,¹ John Pappachan,³ Brian H Cuthbertson,⁴ Laura Drikite,¹ Richard Feltbower,⁵ Theodore Gouliouris,⁶ Isobel Sale,⁷ Robert Shulman,⁸ Lyvonne N Tume,⁹ John Myburgh,¹⁰ Kerry Woolfall,² David A Harrison,¹ Paul R Mouncey,¹ Kathryn Rowan¹ and Nazima Pathan^{11*}

- ¹Clinical Trials Unit, Intensive Care National Audit and Research Centre, London, UK
- ²Department of Public Health, Policy and Systems, University of Liverpool, Liverpool, UK
- ³Paediatric Intensive Care Unit, Southampton Children's Hospital, University of Southampton, Southampton, UK
- ⁴Department of Critical Care Medicine, Sunnybrook Health Sciences Centre, Toronto, Canada
- ⁵PICANet, Division of Epidemiology, University of Leeds, Leeds, UK
- ⁶Clinical Microbiology and Public Health Laboratory, Addenbrooke's Hospital,
- Cambridge University Hospital NHS Foundation Trust, Cambridge, UK ⁷PPI representative, UK
- ⁸Department of Pharmacy, University College London Hospitals NHS Foundation Trust, London, UK
- ⁹Intensive Care Unit, Alder Hey Children's NHS Foundation Trust Liverpool, Liverpool, UK
- ¹⁰George Institute for Global Health, Faculty of Medicine, University of New South Wales, Sydney, Australia
- ¹¹Department of Paediatrics, University of Cambridge, Addenbrooke's Hospital, Cambridge, UK

*Corresponding author np409@cam.ac.uk

Disclosure of interests

Full disclosure of interests: Completed ICMJE forms for all authors, including all related interests, are available in the toolkit on the NIHR Journals Library report publication page at https://doi.org/10.3310/HDKV1008.

Primary conflicts of interest: Nazima Pathan is an Associate Professor of Paediatrics at the University of Cambridge and a member of the National Institute for Health and Care Research (NIHR) Health

Technology Assessment (HTA) research prioritisation panel (March 2021 to March 2026). Kathryn Rowan is the Director of the Clinical Trials Unit at Intensive Care National Audit and Research Centre (ICNARC) and the Programme Director of NIHR Health and Social Care Delivery Research (HSDR) programme and a member of HTA (2007/2009) and HSDR (2016/2019) committees. Lyvonne N Tume is a member of the research prioritisation and funding panels of the NIHR (HTA). John Myburgh is the Chair of Selective Decontamination of the Digestive Tract in the Intensive Care Unit (SuDDICU) Australia Management Committee, Director of The George Institute for Global Health and Leadership Fellowship of National Health and Medical Research Council, Australia. Richard Feltbower is principal investigator at the Paediatric Intensive Care Clinical Audit Network program at Leeds University. David A Harrison is a member of the HTA General Committee.

Disclaimer: This report contains transcripts of interviews conducted in the course of the research, or similar, and contains language which may offend some readers.

Published February 2024 DOI: 10.3310/HDKV1008

Plain language summary

Use of selective gut decontamination in critically ill children: PICnIC a pilot RCT and mixed-methods study

Health Technology Assessment 2024; Vol. 28: No. 8 DOI: 10.3310/HDKV1008

NIHR Journals Library www.journalslibrary.nihr.ac.uk

Plain language summary

Each year, around 20,000 critically ill children are admitted to paediatric intensive care units in the UK. These children are at a higher risk of healthcare-associated infections, one of the main sources of which is the large number of bacteria in the digestive tract. Spread of bacteria from the digestive tract into other organs, such as the lung (causing ventilator-associated pneumonia) or bloodstream (causing sepsis), can be life-threatening. The risk is highest in those children whose illness is so severe that they require prolonged mechanical ventilation.

Stopping the growth of bacteria in the digestive tract (called selective decontamination of the digestive tract) has been shown in adults to reduce the number of hospital-acquired infections. However, there have been no trials in children. We wanted to assess how practical and acceptable such a trial would be comparing standard infection control to selective decontamination of the digestive tract-enhanced infection control and monitoring how each intervention affected antimicrobial resistance.

We undertook a pilot study to examine whether clinicians could identify eligible children, enrol them in the study and follow study procedures during the course of paediatric intensive care unit admission. Alongside this, we interviewed parents and clinicians to get their views on the proposed trial. Six hospitals recruited 559 patients over a period of roughly 7 months. Hospitals were randomly allocated to continue with the standard infection control procedure or to give selective decontamination of the digestive tract. Overall, recruitment was higher than expected. Alongside this, we examined the views of patients, caregivers and healthcare professionals to assess their views on whether a trial should be carried out to see if selective decontamination of the digestive tract should become part of the infection control regime for children most at risk of hospital-acquired infection in the paediatric intensive care unit.

Overall results suggest that a larger PICnIC trial incorporating patient stakeholder and clinical staff feedback on design and outcomes is feasible and that it is appropriate to conduct a trial into the effectiveness of selective decontamination of the digestive tract administration to minimise hospital-acquired infections.

Health Technology Assessment

ISSN 1366-5278 (Print)

ISSN 2046-4924 (Online)

Impact factor: 3.6

A list of Journals Library editors can be found on the NIHR Journals Library website

Launched in 1997, *Health Technology Assessment* (HTA) has an impact factor of 3.6 and is ranked 32nd (out of 105 titles) in the 'Health Care Sciences & Services' category of the Clarivate 2021 Journal Citation Reports (Science Edition). It is also indexed by MEDLINE, CINAHL (EBSCO Information Services, Ipswich, MA, USA), Embase (Elsevier, Amsterdam, the Netherlands), NCBI Bookshelf, DOAJ, Europe PMC, the Cochrane Library (John Wiley & Sons, Inc., Hoboken, NJ, USA), INAHTA, the British Nursing Index (ProQuest LLC, Ann Arbor, MI, USA), Ulrichsweb™ (ProQuest LLC, Ann Arbor, MI, USA) and the Science Citation Index Expanded™ (Clarivate™, Philadelphia, PA, USA).

This journal is a member of and subscribes to the principles of the Committee on Publication Ethics (COPE) (www.publicationethics.org/).

Editorial contact: journals.library@nihr.ac.uk

The full HTA archive is freely available to view online at www.journalslibrary.nihr.ac.uk/hta.

Criteria for inclusion in the Health Technology Assessment journal

Manuscripts are published in *Health Technology Assessment* (HTA) if (1) they have resulted from work for the HTA programme, and (2) they are of a sufficiently high scientific quality as assessed by the reviewers and editors.

Reviews in *Health Technology Assessment* are termed 'systematic' when the account of the search appraisal and synthesis methods (to minimise biases and random errors) would, in theory, permit the replication of the review by others.

HTA programme

Health Technology Assessment (HTA) research is undertaken where some evidence already exists to show that a technology can be effective and this needs to be compared to the current standard intervention to see which works best. Research can evaluate any intervention used in the treatment, prevention or diagnosis of disease, provided the study outcomes lead to findings that have the potential to be of direct benefit to NHS patients. Technologies in this context mean any method used to promote health; prevent and treat disease; and improve rehabilitation or long-term care. They are not confined to new drugs and include any intervention used in the treatment, prevention or diagnosis of disease.

The journal is indexed in NHS Evidence via its abstracts included in MEDLINE and its Technology Assessment Reports inform National Institute for Health and Care Excellence (NICE) guidance. HTA research is also an important source of evidence for National Screening Committee (NSC) policy decisions.

This manuscript

The research reported in this issue of the journal was funded by the HTA programme as award number 16/152/01. The contractual start date was in November 2019. The draft report began editorial review in February 2023 and was accepted for publication in August 2023. The authors have been wholly responsible for all data collection, analysis and interpretation, and for writing up their work. The HTA editors and publisher have tried to ensure the accuracy of the authors' manuscript and would like to thank the reviewers for their constructive comments on the draft document. However, they do not accept liability for damages or losses arising from material published in this manuscript.

This manuscript presents independent research funded by the National Institute for Health and Care Research (NIHR). The views and opinions expressed by authors in this publication are those of the authors and do not necessarily reflect those of the NHS, the NIHR, the HTA programme or the Department of Health and Social Care. If there are verbatim quotations included in this publication the views and opinions expressed by the interviewees are those of the interviewees and do not necessarily reflect those of the NHS, those of the NHS, the NIHR, the HTA programme or the Department of Health and Social Care. If there are verbatim quotations included in this publication the views and opinions expressed by the interviewees are those of the interviewees and do not necessarily reflect those of the authors, those of the NHS, the NIHR, the HTA programme or the Department of Health and Social Care.

Copyright © 2024 Brown *et al.* This work was produced by Brown *et al.* under the terms of a commissioning contract issued by the Secretary of State for Health and Social Care. This is an Open Access publication distributed under the terms of the Creative Commons Attribution CC BY 4.0 licence, which permits unrestricted use, distribution, reproduction and adaptation in any medium and for any purpose provided that it is properly attributed. See: https://creativecommons.org/licenses/by/4.0/. For attribution the title, original author(s), the publication source – NIHR Journals Library, and the DOI of the publication must be cited.

Published by the NIHR Journals Library (www.journalslibrary.nihr.ac.uk), produced by Newgen Digitalworks Pvt Ltd, Chennai, India (www.newgen.co).